Author’s response to reviews

Title: The health and wellbeing needs of veterans: a rapid review

Authors:

Candice Oster (candice.oster@flinders.edu.au)
Andrea Morello (andrea.morello@flinders.edu.au)
Anthony Venning (anthony.venning@flinders.edu.au)
Paula Redpath (paula.redpath@flinders.edu.au)
Sharon Lawn (sharon.lawn@flinders.edu.au)

Version: 1 Date: 06 Sep 2017

Author’s response to reviews:

We would like to thank the reviewer for taking the time to review our revised paper.

Please find below reviewer comments, and our response below.

Yours Sincerely,
The authors

Response to Reviewer Comments

Page 5 Line 7: service may contribute to the development 'OF' mental, physical …

Response: This has been amended.

Figure 1: 386 records identified reduced to 320 after duplicates are removed should mean that 66 duplicates were removed -currently the figure states 86 duplicates were removed

Response: This has been amended.